ELSEVIER

Contents lists available at ScienceDirect

The Veterinary Journal



journal homepage: www.elsevier.com/locate/tvjl

Review

Cartilage defect repair in horses: Current strategies and recent developments in regenerative medicine of the equine joint with emphasis on the surgical approach



Stefan Cokelaere ^{a,*}, Jos Malda ^{a,b}, René van Weeren ^a

^a Department of Equine Sciences, Faculty of Veterinary Medicine, Utrecht University, Yalelaan 112, 3584 CM Utrecht, NL, Netherlands ^b Department of Orthopaedics, University Medical Center Utrecht, Heidelberglaan 100, 3584 CX Utrecht, NL, Netherlands

ARTICLE INFO

Article history: Accepted 7 February 2016

Keywords: Cartilage defect repair Surgical techniques Osteochondral lesions Horse Osteoarthritis

ABSTRACT

Chondral and osteochondral lesions due to injury or other pathology are highly prevalent conditions in horses (and humans) and commonly result in the development of osteoarthritis and progression of joint deterioration. Regenerative medicine of articular cartilage is an emerging clinical treatment option for patients with articular cartilage injury or disease. Functional articular cartilage restoration, however, remains a major challenge, but the field is progressing rapidly and there is an increasing body of supportive clinical and scientific evidence. This review gives an overview of the established and emerging surgical techniques employed for cartilage repair in horses. Through a growing insight in surgical cartilage repair possibilities, surgeons might be more stimulated to explore novel techniques in a clinical setting.

© 2016 Elsevier Ltd. All rights reserved.

Introduction

The principal aim of the treatment of joint problems in the horse focuses on pain relief and minimisation of progression of joint deterioration, thus delaying the onset of osteoarthritis. However, this is palliative only, providing temporary relief but no cure, as the structural integrity of the articular cartilage surface, and potentially the underlying bone, is damaged. In fact, healing or functional repair of osteochondral lesions is still a major challenge in orthopaedic surgery. As early as 1743 William Hunter stated: *"From Hippocrates down to the present age, we shall find, that an ulcerated cartilage is universally allowed to be a very troublesome disease; that it admits of a cure with more difficulty than carious bone; and that, when destroyed, it is not recovered"* (Hunter, 1743) (Fig. 1).

Hyaline cartilage is an avascular, aneural and highly organised tissue that generally forms (inferior) repair tissue. There are three mechanisms for cartilage repair: extrinsic, intrinsic and matrix flow. Extrinsic repair is seen in full-thickness cartilage lesions that extend through the tidemark. Mesenchymal elements from the subchondral bone migrate to the defect and fill it with repair tissue (Hunziker and Rosenberg, 1996; Hunziker, 1999; Hurtig et al., 1988). Small full-thickness lesions of 5 mm or less tend to heal by intrinsic repair (from within the cartilage). This implies proliferation of injured or exposed chondrocytes that produce repair tissue, whereas matrix flow, which relies upon migration of the cartilage margins from the perimeter towards the centre of the defect, plays a role in the repair of all types of lesions, but is most important in smaller defects (Convery et al., 1972; Hurtig et al., 1988). In all cases, the resulting repair tissue consists of type I collagen containing fibrocartilage with inferior structural and functional properties compared to native type II collagen found in hyaline cartilage (Fig. 1). This repair tissue does not integrate well with the surrounding hyaline cartilage and has limited longevity (Mankin, 1974; Hurtig et al., 1988; Desjardins and Hurtig, 1990; Gomoll and Minas, 2014).

Ideally, techniques for cartilage repair should generate a repair tissue with a structure and function approximating native articular cartilage, which integrates well with the adjacent cartilage and bone. Stimulation of the natural repair process is theoretically the best method to achieve this, but it should be realised that for larger defects some form of (temporary) filling, such as implantation of articular grafts, will probably always be necessary.

Regeneration of articular cartilage is challenging in man, but more so in the horse. Equine joints share many features with those of humans (Malda et al., 2012) and the equine model is now well accepted for orthopaedic research (Mcllwraith et al., 2011a). However, in a practical sense clinical conditions are more challenging in the horse whilst economic feasibility is often less. In some joints, like the carpus and fetlock, equine cartilage is thinner, precluding the use of several techniques that would be relevant in humans (Lee et al., 2014). Also, the necessity of immediate weight-bearing postoperatively and the often substantial subchondral bone involvement in the horse needs to be considered before simply applying

^{*} Corresponding author. Tel.: +31625015750. *E-mail address*: S.M.Cokelaere@uu.nl (S. Cokelaere).



Fig. 1. Arthroscopic image of a fibrocartilaginous repaired osteochondral defect in the stifle of an adult horse 1 year after surgical debridement.

techniques used in humans. In a surgical-technical sense, many modern approaches for cartilage repair, like cell and tissue engineered transplantation techniques, add complexity to the surgery, and often preclude the use of arthroscopy.

This review focuses on surgical cartilage repair. While articular cartilage cannot be restored in isolation, the joint should be seen as an organ (Samuels et al., 2008), in which homeostasis plays a crucial role in the success of any intervention. Therefore, supplementary medical care and appropriate rehabilitation are highly likely to be necessary as the whole joint should be treated to achieve the best functional and structural outcome.

Surgical techniques for articular cartilage repair

Surgical approaches to treat joint disorders characterised by tissue damage have been grouped into three categories (Hunziker, 2002; Hunziker et al., 2015): palliative (arthroscopic debridement and lavage), reparative (marrow stimulation techniques) and restorative (osteochondral grafting, autologous chondrocyte implantation or ACI).

Palliative treatments: Cartilage debridement and lavage

Surgical debridement and joint lavage are the first-line palliative treatments, which, together with removal of free bodies in the joint and limited excision of osteophytes, have been shown to decrease pain and improve joint function (McIlwraith et al., 2005).

The goal of debridement of full-thickness cartilage defects is to create a contained lesion by removing loose cartilage and bone together with any other degenerated tissue in order to stimulate spontaneous repair. The depth of the debridement and the angles of the lesion affect the formation of repair tissue in the defect (Rudd et al., 1987). Open or arthroscopically assisted manual curettage gives the best quality of debridement (Drobnic et al., 2010) (Fig. 2a, 2b, 2c). Debridement should be thorough, but not overly aggressive. The amount of residual bone after debridement, especially in the carpus, is an important prognostic factor for return to athletic activity (McIlwraith et al., 1987). Removal of the calcified cartilage layer while retaining the subchondral bone plate provides an optimal effect, whereas leaving the calcified cartilage layer in place markedly restricts healing (Frisbie et al., 2006). In practice, the debridement includes removal of all the dull or yellowish tissue down to the bone, extending laterally to the normal cartilage (Frisbie et al., 2006; Drobnic et al., 2010). To retain congruency with the opposing joint surface, it is advisable to keep as much as possible of the subchondral bone plate intact. The remaining bone in the defect should be viable and crumbly, brownish bone should be removed. Minor bleeding from the subchondral bone can be managed using higher fluid pressure or by local application of epinephrine and thrombin, fibrin sealant or haemostatic sponges (Brittberg, 2008).

An interesting recent observation was the use of hyperosmotic saline irrigation solutions (600 mOsm) that, compared to normal saline (300 mOsm), seemed to be chondroprotective and may help in reducing unintentional cartilage injury during articular reconstructive surgery as well as promoting integrative cartilage repair (Eltawil et al., 2015).

Full-thickness cartilage defects need debriding, but there is debate about partial-thickness defects (Baumgaertner et al., 1990). It is not yet possible to truly restore hyaline cartilage, and consensus favours chondroplasty or trimming of the articular surface with removal of protruding strands or fibrillation by use of mechanical debridement with automated synovial abrasers in cases of partial thickness articular cartilage erosion (Ryan et al., 2003; McIlwraith et al., 2005). Debridement of the articular surface aims to reduce shedding of cartilage-derived debris into the synovial environment. Mechanical debridement or shaving can be frustrating because fibrillation may persist and leave behind unviable cartilage or the iatrogenic removal of healthy cartilage may occur (Hunziker and Quin, 2003). Thermal chondroplasty is another technique for smoothing the articular surface. This needs to be used with caution, however,



Fig. 2. Debridement of an osteochondral lesion in the fetlock of an adult horse: (a) severe cartilage (subchondral bone) damage on the sagittal ridge of the third metacarpal bone, (b) curettage with removal of the abnormal cartilage/subchondral bone, and (c) final version of the debrided osteochondral defect.

asexcessive matrix debridement or chondrocyte death can easily result (Ryan et al., 2003; Lubowitz, 2006).

In osteochondritis dissecans (OCD) arthroscopic debridement of lesions in the femoropatellar joint is recommended in horses >11 months of age where lesions are more than 2 cm in length or >5 mm in depth. In the tarsocrural joint, if clinical signs are present, surgery is recommended in all predilection sites, preferably after 5 months of age (McIlwraith, 2013). The overall success rate of surgical treatment of OCD lesions at the distal intermediate ridge of the tibia varied from 74 to 87% (McIlwraith et al., 1991; McIlwraith, 2013). In a recent retrospective study on outcome of arthroscopic debridement of OCD lesions on the lateral femoral trochlear ridge in mainly Warmbloods, 65% of horses returned to their intended use, which was comparable with previous studies (Foland et al., 1992; McIlwraith, 2013). The length of the lesion was significantly associated with the prognosis in horses with lesions <2 cm in length having 78% return to function, against 54% for lesions >4 cm (McIlwraith, 2013; UpRichard et al., 2013). Also in subchondral cystic lesions of the medial femoral condyle the amount of cartilage surface damage is an important prognostic factor. In a study of 150 cases, 70% of horses with <15 mm of surface debridement returned to their previous use compared with 30% of horses with >15 mm defects (Sandler et al., 2002). In another study, breed differences were reported with Quarter horses having poorer results after debridement compared with Thoroughbreds and Arabians (Howard et al., 1995). There is also an age effect. After arthroscopic debridement of a subchondral cystic lesion of the medial femoral condyle 64% of horses aged 0-3 years returned to soundness, against only 34% of horses older than 3 years (Smith et al., 2005). Retention of the cartilage surface over the cystic lesions is important (Sandler et al., 2002), and there is an association between debridement of medial femoral condylar defects and medial meniscus or meniscal ligamentous injury (Hendrix et al., 2010).

Critically, debridement of full-thickness cartilage lesions results in fibrocartilage and imperfect hyaline repair tissue. A permanent irregularity of the subchondral bone plate frequently persists, as has been demonstrated on long-term follow-up radiographs (McIlwraith et al., 2014). Especially in cases of larger lesions, such as advanced OCD cases of the stifle, development of osteoarthritis and persistent lameness may ensue. While debridement remains the first step to stimulate and start cartilage repair, additional reparative procedures should thus be considered.

Reparative techniques: Bone marrow stimulation techniques (BSTs)

Bone marrow stimulation techniques are among the first-line surgical treatment options for symptomatic articular cartilage defects. BSTs are based on the penetration of the subchondral bone plate at the base of the cartilage defect, evoking an endogenous repair response. Bleeding from the bone marrow under the subchondral plate will bring a low number of pluripotent stem cells and growth factors to the defect that are embedded in the blood-clot that forms at the base of the defect and are crucial for the early stages of repair (Frisbie et al., 1999). The procedure is also said to result in improved anchorage of the repair tissue to the underlying subchondral bone and to some extent to the surrounding cartilage (Steadman et al., 2010).

Surgically, penetration of the subchondral bone spaces can be accomplished with small drill bits (subchondral drilling), awls (microfracture) or more aggressively by generalised or limited abrasion with burs (abrasion arthroplasty). Abrasion arthroplasty involves removal of a uniform layer of residual cartilage and a superficial layer of dead subchondral bone until mild bleeding is induced. Positive clinical results have been reported for OC lesions in the hock and



Fig. 3. Microfracturing (with an awl) of an osteochondral defect after debridement in the stifle of an adult horse.

stifle, but no comparative studies have been published (Scott et al., 2004). A more aggressive version of abrasion arthroplasty is called spongialisation and involves complete removal of the subchondral bone plate. Although this may be beneficial in cases with advanced cartilage and subchondral bone damage (Ficat et al., 1979), it is preferable to preserve the integrity of the subchondral bone plate and limit the use of this technique to exceptional cases.

Subchondral drilling or forage was proposed in humans for the treatment of osteochondrosis dissecans and for knee osteoarthritis (Smillie, 1957; Pridie, 1959). Controlled horse studies using subchondral drilling (1-mm diameter 1-cm deep holes) showed better healing compared to untreated defects in partial-thickness defects in the third carpal bone (Shamis et al., 1989). Another study revealed a greater surface coverage with more fibrocartilaginous tissue and better attachment to the underlying tissue (Vachon et al., 1986). Subchondral drilling also has been used in the sclerotic rim of subchondral cystic lesions, but led to complications such as cyst enlargement or development of additional cysts or intralesional osteophytes (Howard et al., 1995; McIlwraith et al., 2005).

Arthroscopic microfracturing or micropicking of lesions in humans, aiming at recruitment of MSCs, was introduced in the early 1980s (Steadman et al., 1997) (Fig. 3). This technique is simple, cost-effective and forms a basis for further treatments. Before microfracturing, removal of the calcified cartilage while retaining the integrity of the subchondral bone plate is necessary. Through various handheld angled tapered awls or picks, microfractures are made in the subchondral bone perpendicular to the surface, starting at the periphery and progressing towards the centre of the defect. Microfracture holes are placed 3-5 mm apart to cover the entire debrided area in a cartilage defect, including the perimeter, as this encourages tissue healing at the junction of the residual cartilage and the new repair tissue. The microfracture awls should penetrate 2 to 4 mm into the subchondral bone to optimise the access to the bone marrow. The rough surface that is created facilitates anchorage of the repair tissue.

Clinical trial results of microfracture in equine joint disease have not been published yet, but several experimental studies have been described. Frisbie et al. (1999) showed improvement in the quantity of repair tissue and type II collagen content and earlier bone remodelling at 4 and 12 months after microfracture of full-thickness defects through the calcified cartilage, but not penetrating the subchondral bone plate. The repair tissue was characterised superficially by fibrous tissue with deeper a mixture of fibrocartilage and hyaline cartilage. The difference between treated and control sites had been established at or before 4 months with no change between 4 and 12 months. At 12 months, degeneration in repair tissue was noted in the femoral condyles compared to the radial carpal bones, probably due to biomechanical instability caused by enzymatic degradation of the repair tissue. In this and another study it appeared that thorough and complete removal of the calcified cartilage improves overall repair tissue attachment (Frisbie et al., 1999, 2006). In a further study type II collagen gene expression was increased within 8 weeks of microfracture, whereas aggrecan expression appeared uninfluenced (Frisbie et al., 2003).

The microfracture technique does however also have some drawbacks compared to drilling. In a mature rabbit model, cartilage repair was impaired due to compaction and fracturing of bone around microfracture holes compared to drilling with cooled irrigation that permitted clearance of bone from the holes with improved access to the bone marrow. Shearing and crushing of adjacent bone with microfracture caused more osteocyte death than heat by drilling (Chen et al., 2009). In a rabbit study on the effect of microfracture and drill hole depth (6 mm vs. 2 mm) and hole type (microfracture versus drill), deeper drilling was more beneficial for cartilage repair and no difference in quantity and quality of repair was observed between microfracture and drilling to 2 mm depth (Chen et al., 2011). However, subchondral drilling caused longer lasting damage to the subchondral microarchitecture, with more formation of subchondral cysts and intralesional osteophytes in a sheep model (Orth et al., 2012). Use of smaller drill holes induces less subchondral bone disturbance and is more effective for osteochondral repair (Eldracher et al., 2014). A sheep study showed that the use of small-diameter awls of 1 mm instead of larger awls decreased the extent of subchondral bone damage and improved articular cartilage repair (Orth et al., 2015). Nanofracturing or subchondral needling represents an innovative technique that overcomes the shortcomings of microfracturing with a better bone marrow access (Benthien and Behrens, 2015; Zedde et al., 2015).

Microfracture studies in humans have shown that the repair tissue is a form of fibrocartilage at best and that the clinical outcome deteriorates after 2 years of follow-up, with inconclusive durability and possible treatment failure beyond 5 years (Kreuz et al., 2006; Mithoefer et al., 2009; Goyal et al., 2013). Subchondral bone alterations such as subchondral cystic lesions or intralesional osteophytes were also seen in up to one-third of patients treated with microfracture (Kreuz et al., 2006; Mithoefer et al., 2009).

Microfracture can be performed alone, or in combination with other reparative procedures to improve healing. Autologous matrixinduced chondrogenesis (AMIC) is single-stage technique that is a combination of microfracture and a scaffold or biomembrane that is sutured or glued into the cartilage defect. The implanted matrix is thought to stabilise the resulting blood clot and prevent loss of cells (Benthien and Behrens, 2011). Enhanced marrow stimulation and formation of more hyaline-like cartilage have been achieved when using scaffolds like collagen type I/III porcine membrane (Chondrogide) or a chitosan-based liquid scaffold (BST-Cargel) (Benthien and Behrens, 2011; Stanish et al., 2013; Bark et al., 2014). Further biologic enhancement of microfracturing or AMIC in human and animal studies has been achieved by delivery of growth factors (platelet rich plasma, autologous serum), hyaluronic acid or even pulse electro-magnetic fields (Mirza et al., 2015; McIlwraith et al., 2011b; Doral et al., 2012; Makris et al., 2015, Karakaplan et al., 2015; Marmotti et al., 2015).

The field of tissue regeneration and marrow stimulation is constantly evolving. Further well-designed studies in horses (and humans) are needed to determine the long-term efficacy and define the usefulness of each technique and its specific clinical indications compared to other cartilage repair techniques (Bert, 2015; Makris et al., 2015).

Restorative techniques: Transplantation procedures (grafting)

Tissue-based transplantation procedures

Cartilage reattachment

Salvage and reattachment of large OCD flap lesion using multiple resorbable polydioxanone pins (>2 cm) has been described by Nixon et al. (2004). The OCD flap should be still in situ within the original defect, with some residual continuity along at least 50% of the perimeter with normal surrounding cartilage. The surface of the OCD flap must be smooth and congruent with minimal fibrillation and the flap should not be entirely mineralised. The method leads to rapid resolution of joint effusion and lameness and the reconstitution of the subchondral surface. In a study in 27 horses, long-term follow-up with a mean duration of 15.6 months revealed an overall success rate for intended athletic use of 95%. Radiographic improvement started within weeks of surgery and many lesions resolved radiographically within 3 to 6 months of surgery with better reformation of the subchondral bone contour than following cartilage flap removal (Sparks et al., 2011b).

Subchondral cystic lesions (type 1) of the medial femoral condyle in three horses have been treated successfully by polydioxanone pin reattachment of the overlying cartilage and injection of the cystic cavity with bone marrow aspirate concentrate or allogeneic chondrocytes (Sparks et al., 2011a).

Osteochondral autografting and allografting

Larger defects that involve subchondral bone require osteochondral transplantation. Osteochondral autografts and allografts have the advantage of immediate reconstruction of the articular surface by transfer of mature intact hyaline cartilage and the underlying subchondral bone (Hangody and Fules, 2003). These resurfacing techniques have been employed successfully in humans since the mid-1990s (Hangody et al., 1997). The success is mainly dependent on graft chondrocyte viability and mechanical stability of the host– graft interface (Pallante et al., 2012a).

Osteochondral autologous transfer (OAT) and mosaicplasty focus on transferring one large cylindrical osteochondral graft (dowel/ shell graft) or multiple smaller cylindrical grafts from minimally weight-bearing regions in regions with greater weight-bearing (Lynch et al., 2015). Mosaicplasty, which uses a number of small-diameter cylindrical heterotopic osteochondral grafts placed in mosaic-like fashion, has gained popularity, also in horses, as it enables the relatively easy reconstruction of a relatively congruous joint surface (Hangody and Fules, 2003; Bodo et al., 2014). However, donor site availability and joint congruency remain major concerns, limiting the use of osteochondral autograft transfer. Allografts, harvested from available cadaveric donors, overcome these limitations and have the advantage of orthotopic transfer of the harvested tissue (Bugbee et al., 2015; De Caro et al., 2015).

In humans, a fresh osteochondral allograft transplantation (mega-OAT) can be used in large osteochondral defects or if another cartilage repair procedure has failed (Sherman et al., 2014). Fresh osteochondral allografts stored at 37 °C or room temperature have the highest chondrocyte viability (Pallante et al., 2012b). Host response to the allogenous tissue, mainly to the bone, can affect cartilage viability. As chondrocytes are tightly embedded in the extracellular matrix, immunogenicity seems to be low and seems to have minimal impact on graft survival (Hurtig et al., 2001; Bugbee et al., 2015; Smith et al., 2015). Currently, shell osteochondral allografts have the subchondral bone plate as the only bony component that allows secure graft fixation while minimising potential immunogenicity (Smith et al., 2015).

Long-term follow-up (up to 10 years) in humans shows good clinical and functional outcomes with osteochondral grafting (Hangody and Fules, 2003; De Caro et al., 2015; Lynch et al., 2015).

Mosaic arthroplasty has been used successfully clinically and experimentally in horses with subchondral cystic lesions in the stifle, fetlock and tarsal joints (Bodo et al., 2000, 2004, 2014; Janicek et al., 2010). Osteochondral grafts are harvested from a less weightbearing region and transferred to a prepared cylindrical hole into the defect, using arthrotomy and/or arthroscopy. Good matching of donor and recipient sites with respect to cartilage thickness, biochemical and physical properties is necessary. In a stifle matching study, grafts from the trochlear groove and axial aspect of the lateral trochlear ridge were the closest match for the medial femoral condyle, whereas grafts from the trochlear groove and axial aspect of the medial trochlear ridge matched closest with the lateral femoral condyle (Changoor et al., 2006). Mosaic arthroplasty of other locations, like the third carpal bone, yielded mixed results. In those locations, tissue matching was not possible, hence the transplants needed to adapt to their new biomechanical environment (Hurtig et al., 2001). Overall, the use of the technique in horses is limited due to the technical difficulties in harvesting and insertion, especially when using arthroscopy.

Particulated cartilage grafts

Particulated autografts and off-the-shelf allografts are constructs on the basis of minced cartilage placed in a scaffold or fibrin glue (Albrecht et al., 1983; Makris et al., 2015). Mechanical mincing allows chondrocytes to escape from their surrounding extracellular matrix, migrate to surrounding tissues and form a new hyalinelike cartilage tissue matrix (Albrecht et al., 1983; Riboh et al., 2015). The procedure is easy to perform and can be done in a minimally invasive single-stage procedure.

The cartilage autograft implantation system (CAIS) involves mechanical fragmentation of arthroscopically harvested cartilage from a low-load-bearing region. Cartilage fragments are subsequently placed on a synthetic, absorbable scaffold using fibrin glue and fixed with synthetic, absorbable staples into the defect (without microfracture). Good long-term (12 months) healing has been demonstrated in an equine model (Frisbie et al., 2009). Based on the same principles, minced allografts of juvenile articular cartilage have been used, as the gene expression pattern in juvenile cartilage is more favourable for regeneration and there is increased proliferative potential in juvenile compared to adult chondrocytes (Riboh et al., 2015; Yanke et al., 2015).

BioCartilage is a dehydrated micronised allograft cartilage scaffold that is mixed into a paste with platelet-rich plasma (PRP) implanted over a microfractured defect. Studies in humans reported more hyaline-like tissue compared to microfracture alone (Abrams et al., 2013). In a recent equine study comparing BioCartilage to microfracture alone, better repair-host integration, base integration and collagen type II 13 months postoperatively were observed in chondral defects treated with BioCartilage compared to microfracture alone (Cole et al., 2015).

Cell-based transplantation procedures

Chondrocyte-based strategies

Autologous chondrocyte implantation (ACI) was first reported in humans in 1994 for the treatment of focal chondral injury in the tibiofemoral and patellofemoral joints (Brittberg et al., 1994). Clinical outcome in long-term human studies has been reported as good to excellent for treatment of symptomatic full-thickness chondral or osteochondral lesions in the femoral condyle, with 80% to 90% return to pain-free function (Brittberg, 2008; Peterson et al., 2010). In the original first-generation ACI-technique, a small biopsy of healthy articular cartilage was arthroscopically harvested from a lowweight-bearing location of the diseased joint. A few hundred thousands of chondrocytes were isolated, expanded to generate more than 10 million cells and subsequently re-implanted into the defect, which was then covered by a periosteal flap sutured to the margins (Brittberg et al., 1994). Disadvantages of the technique are that the delivery of the cells requires an arthrotomy and harvesting and placement of a periosteal flap is technically demanding, with postoperative graft delamination, tissue hypertrophy, graft failure and adhesions as possible complications. In subsequent modifications of the technique, collagen-membranes have been used instead of a periosteal flap (second generation ACI), as well as cell-delivery in a variety of three-dimensional scaffolds with different methods of fixation (third generation ACI or MACI for matrix-induced autologous chondrocyte implantation) (Kon et al., 2013; Gobbi et al., 2015).

A number of studies have been carried out in horses in the past two decades. Initial trials of arthroscopically resurfacing 12 mm articular cartilage defects with allogeneic chondrocytes secured in a fibrin matrix showed an improved filling of the defect with significantly more aggrecan and 62% more type II collagen at 8 months, compared to empty defects (Hendrickson et al., 1994). As fibrin is difficult to attach in the graft site, methods using tissue-engineering approaches with collagen scaffolds were developed, but did not provide satisfactory improvement in repair of 15 mm cartilage defects (Sams and Nixon, 1995; Sams et al., 1995). Addition of anabolic growth factors (especially IGF-1) showed enhanced cartilage repair through the stimulation of matrix synthesis by the transplanted chondrocytes, resulting in substantially improved histologic appearance after 8 months, over controls. However, biochemical and biomechanical characteristics of the repair tissue were not significantly improved (Fortier et al., 2002). A retrospective study on treatment of subchondral cystic lesions of the medial femoral condyle in 49 mature horses with this combination showed a successful longterm clinical outcome in 74% of cases (Ortved et al., 2012). Articular chondrocytes genetically modified to express high levels of IGF-1 or bone morphogenetic protein (BMP)-7 have also been shown to improve the quality of repair tissue (Hidaka et al., 2003; Ortved et al., 2014). ACI has been proven effective for both short- and longterm repair of large partial-thickness and full-thickness cartilage defects in horses (Litzke et al., 2004; Nixon et al., 2011). However, due to the complexity of the whole procedure, the different approaches using ACI seem to be unfeasible or at least very impractical for the majority of equine surgical practices. Concern about uneven distribution of the cells in the defect and cell-leakage with injection of expanded chondrocyte suspension underneath a membrane led to further improvement of the ACI technique (Berta et al., 2015; Makris et al., 2015).

Matrix-induced autologous chondrocyte implantation (MACI) is a scaffold-plus-cell-based cartilage repair technique and has been performed in a few controlled horse studies as well. Results at 12 and 18 months of a modified MACI technique with autologous chondrocytes cultured on a collagen membrane implanted with PDS/ PGA staples into 15 mm defects of the medial trochlear ridge of the femur showed effective repair compared with the collagen membrane alone and to empty defects (Frisbie et al., 2008). A recent 6-month study in horses using the original MACI implant, secured with fibrin into 15 mm femoral trochlear ridge defects, showed MACI yielded improved arthroscopic second-look, gross healing, and composite histologic scores, compared to spontaneously healing empty defects (Nixon et al., 2015). Biomechanically, the neo-tissue had similar compressive and frictional properties to native tissue, with inferior shear properties (Griffin et al., 2015).

With respect to cell source, the use of autologous versus allogeneic cells for transplantation is still under debate, related to risks of OA development due to the biopsy or immunologic reactions respectively (Hunziker et al., 2015). Donor chondrocyte viability is important for the success of chondrocyte transplantation and when using allogeneic cells, development of a banked source of equine chondrocytes is an integral part of the approach. Foals of 3 to 12 months old and young adult horses are better donors than neonatal foals (Nixon et al., 1992). Results of an in vitro study using rat cells support the potential use of allogeneic chondrocytes in OA and cartilage defects. In this study, the lack of evident immunogenicity, despite exposure to a pro-inflammatory environment, coupled with the immunomodulatory ability suggested that these cells have the potential to evade the host immune system and suppress inflammation, thus potentially facilitating the resolution of OA induced inflammation and cartilage regeneration (Lohan et al., 2015).

Stem cell-based strategies

As chondrocyte based-strategies have been hampered by difficulties in achieving a high cell density and in maintaining their differentiation state, chondroprogenitor cells and mesenchymal stromal/stem cells (MSCs) have been shown to be a potential alternative cell source (Fortier et al., 1998; McCarthy et al., 2012; Frisbie et al., 2015; Madeira et al., 2015). Several commercial stem cellbased therapeutic options to treat joint disease are now available for equine practitioners. Currently bone marrow (BM-MSCs) or bone marrow concentrate (BMC) and adipose tissue (AT-MSCs) are the main sources of MSCs (Gutierrez-Nibeyro, 2011). Peripheral blood, umbilical cord blood, induced pluripotent stem cells (iPSCs) and synovial fluid and membrane are other potential chondrogenic cell sources (Koch et al., 2007; Berg et al., 2009; De Schauwer et al., 2011; Klontzas et al., 2015; Makris et al., 2015; Prado et al., 2015; Spaas et al., 2015; Williams et al., 2015).

Bone marrow aspirate is harvested from the sternum or ilium and adipose tissue is generally harvested from the tail head region (Taylor and Clegg, 2011; Adams et al., 2012) with aspirates from the ilium producing significantly better results than other sources in terms of tissue production, when matrix production of the cells is assessed following chondrogenic differentiation (Kisiday et al., 2013). Long-term cryopreservation of MSCs is possible, without loss of their proliferative and differentiation potential (Marquez-Curtis et al., 2015). MSCs can be grafted into a lesion using a scaffold, injected as bone marrow concentrate, or injected intra-articularly either alone or in combination with other products such as hyaluronan or plateletrich plasma (Schnabel et al., 2013). A technique using a fibrin scaffold to retain MSCs within cartilage defects has been developed. Disadvantages with this technique are the need for preparation of autologous MSCs and the requirement of specialised equipment for arthroscopic implantation of MSC scaffolds (Frisbie and Stewart, 2011). The debate on the best cell type to be used is still ongoing. It remains unclear whether equine allogeneic MSCs incite an immune response, especially if administered repeatedly (Schnabel et al., 2013). Pigott et al. (2013) showed only a moderate inflammatory response with intra-articular injection, while Pezzanite et al. (2015) showed strong in vivo antibody responses to allogeneic BM-MSCs and cross-reactivity with MHC-types other than that of the donor. This can limit the clinical effectiveness of repeated injections, apart from generating untoward inflammatory responses.

To date, the published literature on the outcome of cartilage repair using MSCs in the horse is limited. Efficacy for the treatment of equine osteoarthritis (OA) and cartilage injuries seems more favourable for bone marrow-derived cells than adipose-derived cells (Wilke et al., 2007; Frisbie et al., 2009; Fortier et al., 2010; Ferris et al., 2014). BM-MSCs fibrin grafts in a 15 mm full-thickness cartilage defect model in the lateral femoral trochlear ridge improved the early healing response, but did not significantly enhance the longterm histologic appearance or biochemical composition (Wilke et al., 2007). A significant improvement was demonstrated in macroscopic and histologic scoring at 3 and 8 months for the defects treated by microfracture and grafted with BMC compared to microfracture alone in a similar model (Fortier et al., 2010). Delayed direct intraarticular injection of BM-MSCs in hyaluronan (HA) for the treatment of microfractured full-thickness medial femoral condylar cartilage defects led to increased firmness of repair tissue at 12 months, as well as increased aggrecan content (McIlwraith et al., 2011a). In a clinical follow-up of 33 horses with stifle injuries receiving an intraarticular injection of BM-MSCs following arthroscopic surgery, Ferris et al. (2014) showed clinical improvement with better ability to return to work compared to surgery alone and improved results for horses with meniscal injuries; joint flare post injection was 9%. An equine cartilage explant study demonstrated that chondrogenic priming of allogeneic peripheral blood MSCs (PB-MSCs) resulted in significantly enhanced and homogenous MSC adhesion and incorporation into lesions compared to unprimed cells. Interestingly, mechanical loading negatively affected the results and a lower stem cell density (0.5×10^6) yielded better results than a higher one (1.0×10^6) (Spaas et al., 2015).

Use of high numbers of stem cells alone for regenerative purposes can provoke unwanted side-effects. Formation of significant amounts of bone in the repair tissue may result when using an autologous platelet enhanced fibrin (APEF) scaffold combined with BM-MSCs (Goodrich et al., 2013). In the study by Fortier et al. (2010) this effect was not observed with bone marrow concentrate, which contained a much lower number of MSCs.

Articular cartilage progenitor cells, present in developing and adult normal and osteoarthritic articular cartilage, synovium and adipose tissue, demonstrate a multipotent differentiation capacity similar to that of BM-MSCs (Jayasuriya and Chen, 2015). In contrast with BM-MSCs, these cells retain their chondrogenic potential following extensive in vitro expansion and have limited risk for repair tissue mineralisation in vivo (McCarthy et al., 2012; Frisbie et al., 2015).

Co-culture of chondrocytes or even peripheral blood derived mononuclear cells (PBMCs) and stem cells is known to enhance MSC migration, chondrogenic differentiation and cartilage matrix production and seems to have greater promise for cartilage repair than MSCs alone (Wu et al., 2011; De Windt et al., 2015; Hopper et al., 2015a,b). PBMCs can acquire an MSC-like phenotype and have also shown to attract native chondrocytes from the diseased tissue to aid in cartilage repair (Hopper et al., 2015c). The therapeutic application of stem cells in joint disease holds great promise, but there remain many unanswered questions, warranting experimental and clinical studies.

Tissue-engineering strategies: The use of seeded/unseeded scaffolds of various types and 3D-bioprinting

Although the previously described techniques, like ACI and MACI, improve clinical outcome, they are not the definitive answer for (osteo-)chondral defect repair. None of the techniques result in the restoration of hyaline cartilage that is functionally equal to the original tissue and hence long-term prognosis is still guarded. There is thus an ongoing quest for novel techniques that are more efficacious. Tissue-engineering approaches are potentially promising, as they allow avoidance of donor site morbidity, host immune responses, and disease transmission (Seo et al., 2015b; Smith and Grande, 2015). Cells, 3D-scaffolds and growth factors are the three main components in tissue-engineering. Scaffolds, in different physical forms like fibres, meshes and gels, can be derived from biological or synthetic materials or combinations and can be functionalised with growth factors (such as BMP-2, TGF- β 1, PRP) and/or mimicking peptides (Barnewitz et al., 2006; Seo et al., 2015a). Scaffolds can be used seeded with chondrocytes or stem/progenitor cells or unseeded, as scaffolds are not only a carrier system, but may also have an intrinsic regenerative ability (Kon et al., 2015; Seo et al. 2015b). They temporarily fill the defect and are ideally degraded and replaced simultaneously in a synchronised process by neo-tissue (Seo et al. 2015b; Smith and Grande, 2015).



Fig. 4. Perioperative view of an implanted decellularised scaffold in an osteochondral defect (11 mm \times 10 mm) of the medial femoral trochlear ridge in an adult horse (experimental study).

Scaffolds may be based on the natural extracellular matrix (ECM) of a variety of tissues, which are decellularised (Fig. 4) and otherwise processed before being implanted. These scaffolds are thought to be well prepared for tissue engineering, as their natural ECM provides a unique, tissue-specific 3D environment containing both structural and functional molecules which, in interaction with the resident cells, determine tissue homeostasis (Benders et al., 2013). Chondrogenic potency of multipotent stromal cells, more than chondrocytes, on equine cartilage-derived matrix (CDM) scaffolds has been demonstrated in vitro (Benders et al., 2014). However, there are several issues that still need to be addressed when using cartilage tissue-engineering. These questions consist of (1) the recapitulation of the zonal structure of natural cartilage by these scaffolds; (2) which tissue might form the ideal basis for this type of scaffolds (collagen type II vs. type I); and (3) how to generate a neo-tissue with biphasic character (cartilage and bone), while minimising the risk of overgrowth of the latter (Klein et al., 2009).

Artificial scaffolds have the advantage that the composition is more reproducible, that immune-related problems are less likely to occur, and that regulatory approval will be easier and quicker to obtain. Recently, the use of artificial scaffolds seeded with extracellular vesicles rather than cells has been proposed, which is an exciting option as the biological effect may mimic that of cells whilst avoiding the regulatory problems associated with the use of living (allogeneic) cells (Malda et al., 2016). All scaffold-based approaches need extensive *in vitro* testing and, depending on their concept, functionalisation may require adding extra components, such as enzymes, cytokines and growth factors, that can already naturally be present in ECM.

Hydrogels prepared from natural and synthetic polymers are usually the principal component of scaffolds, as they are biocompatible and can exhibit comparable swelling and lubricating behaviour as articular cartilage, whilst offering an excellent environment for chondrocytes or stem cells (Spiller et al., 2011; Martins et al., 2014; Seo et al., 2015b; Vilela et al., 2015). Several types of hydrogels have been used for cartilage tissue engineering or are being developed (Schuurman et al., 2013; Levett et al., 2014; Moreira Teixeira et al., 2014). A great disadvantage of hydrogels is their low intrinsic stiffness, which is far below the stiffness of natural cartilage. There are various ways to address this problem, such as photo crosslinking, the use of chemicals, or the fabrication of a hybrid scaffold in which the hydrogel is combined with much stiffer materials such as $poly(\epsilon)caprolactone$ (PCL) or other fibrous materials (Schuurman et al., 2011; Wang et al., 2014; Seo et al., 2015a; Visser et al., 2015). Another item is fine-tuning of the in vivo degradation rate of the material, allowing for the gradual and simultaneous replacement of the scaffold by newly formed repair tissue.

Biofabrication is a rapidly developing field in regenerative medicine that holds promise for tissue engineering of osteochondral defects (Di Bella et al., 2015; Groll et al., 2016). In particular three-dimensional bioprinting, one of the main approaches in biofabrication, allows for the fabrication of biphasic, i.e. osseous and cartilaginous, constructs through the manipulation of scaffold composition and cell type (Levato et al., 2014; Seo et al., 2015b). It also allows for the mimicking of the zonal structure of cartilage and potentially larger anatomical structures (Visser et al., 2013). The research is slowly passing from the *in vitro* phase and the testing in small laboratory species to application in large animal models, of which the horse has been designed as one of the best models for cartilage repair (McIlwraith et al., 2011a). Significant breakthroughs are to be expected in the coming years.

Conclusions

This review focuses on the current and future (surgical) cartilage regeneration strategies with their advantages, limitations and outcomes. Simple debridement and marrow stimulation techniques remain the mainstay in equine joint surgery. These techniques are simple and cost-effective, but do not result in the regeneration of hyaline cartilage. More advanced techniques aiming at improved anatomic and functional restoration have been used in horses, often in an experimental setting. These include cartilage reattachment, osteochondral grafting, chondrocyte-based and stem cellbased strategies. Outcomes in the short or medium term have been favourable in some cases, but the complexity and sometimes the invasive nature of those procedures, together with the high costs, are serious drawbacks for routine clinical use. Long-term functional outcome, which is essential in the equine industry, is as yet uncertain and many of the described techniques will still result in a more or less inferior fibrocartilaginous repair tissue and subchondral bone changes at the defect site, as is the case with the currently clinically used approaches. For this reason, there is still a large and thus far unmet need to develop novel tissue engineering techniques and to address the fundamental requirements of successful cartilage healing and joint function, in horses as well as in humans. To eventually reach this goal, well-designed, prospective, randomised controlled (pre-)clinical trials are crucial to compare novel technologies to current 'gold standard' clinical approaches. In this way, treatment algorithms can be formulated for horses, as in humans, to help guide the decision to treat the cartilage defect in the best possible way. The harsh biomechanical environment of the equine joint is a complicating factor that should be taken into account. Additionally the design of effective rehabilitation protocols is paramount, as these will certainly affect outcome of any cartilage repair strategy. The ultimate goal of achieving functional and durable healing of articular cartilage may still seem a long way off, but progress is certainly being made and we are getting closer in making this concept into a clinical reality.

Conflict of interest statement

None of the authors of this paper has a financial or personal relationship with other people or organisations that could inappropriately influence or bias the content of the paper.

Acknowledgements

The authors gratefully acknowledge the financial support of the Dutch Arthritis Foundation (Project LLP 22) and the European Community's Seventh Framework Programme (FP7/2007-2013) under grant agreement 309962 (HydroZONES).

References

- Abrams, G.D., Mall, N.A., Fortier, L.A., Roller, B.L., Cole, J.B., 2013. BioCartilage: Background and operative technique. Operative Techniques in Sports Medicine 21, 116–124.
- Adams, M.K., Goodrich, L.R., Rao, S., Olea-Popelka, F., Philips, N., Kisiday, J.D., McIlwraith, C.W., 2012. Equine bone marrow-derived mesenchymal stromal cells (BMDMSCs) from the ilium and sternum: Are there differences? Equine Veterinary Journal 45, 372–375.
- Albrecht, F., Roessner, A., Zimmerman, E., 1983. Closure of osteochondral lesions using chondral fragments and fibrin adhesive. Archives of Orthopaedic and Trauma Surgery 101, 213–217.
- Bark, S., Piontek, T., Behrens, P., Mkalaluh, S., Varoga, D., Gille, J., 2014. Enhanced microfracture techniques in cartilage knee surgery: Fact or fiction? World Journal of Orthopedics 18, 444–449.
- Barnewitz, D., Endres, M., Krüger, I., Becker, A., Zimmerman, J., Wilke, I., Ringe, J., Sittinger, M., Kaps, C., 2006. Treatment of articular cartilage defects in horses with polymer-based cartilage tissue engineering grafts. Biomaterials 27, 2882– 2889.
- Baumgaertner, M.B., Cannon, W.D., Vittori, J.M., Schmidt, E.S., Maurer, R.C., 1990. Arthroscopic debridement of the arthritic knee. Clinical Orthopaedics and Related Research 253, 197–202.
- Benders, K.E., van Weeren, P.R., Badylak, S.F., Saris, D.B., Dhert, W.J., Malda, J., 2013. Extracellular matrix scaffolds for cartilage and bone regeneration. Trends in Biotechnology 31, 169–176.
- Benders, K.E., Boot, W., Cokelaere, S.M., van Weeren, P.R., Gawlitta, D., Bergman, H.J., Saris, D.B.F., Dhert, W.J.A., Malda, J., 2014. Multipotent stromal cells outperform chondrocytes on cartilage-derived matrix scaffolds. Cartilage 5, 221–230.
- Benthien, J.P., Behrens, P., 2011. The treatment of chondral and osteochondral defects of the knee with autologous matrix-induced-chondrogenesis (AMIC): Method description and recent developments. Knee Surgery, Sports Traumatology, Arthroscopy 19, 1316–1319.
- Benthien, J.P., Behrens, P., 2015. Nanofractured autologous matrix induced chondrogenesis (NAMIC©) – Further development of collagen membrane aided chondrogenesis combined with subchondral needling: A technical note. The Knee 22, 411–415.
- Berg, L.C., Koch, T.G., Heerkens, T., Besonov, K., Thomsen, P.D., Betts, D.H., 2009. Chondrogenic potential of mesenchymal stromal cells derived from equine bone marrow and umbilical cord blood. Veterinary Comparative Orthopaedics and Traumatology 22, 363–370.
- Bert, J.M., 2015. Abandoning microfracture of the knee: Has the time come? Arthroscopy: The Journal of Arthroscopic and Related Surgery 31, 501–505.
- Berta, A., Duska, Z., Toth, F., Hangody, L., 2015. Clinical experiences with cartilage repair techniques: Outcomes, indications, contraindications and rehabilitations. Eklem Hastalik Cerrahisi 26, 84–96.
- Bodo, G., Hangody, L., Szabo, Z., Peham, C., Schinzel, M., Girtler, D., Sotonyi, P., 2000. Arthroscopic autologous osteochondral mosaicplasty for the treatment of subchondral cystic lesion in the medial femoral condyle in a horse. Acta Veterinaria Hungarica 48, 343–354.
- Bodo, G., Hangody, L., Modis, L., Hurtig, M., 2004. Autologous osteochondral grafting (mosaic arthroplasty) for treatment of subchondral cystic lesions in the equine stifle and fetlock joints. Veterinary Surgery 33, 588–596.
- Bodo, G., Vasarhelyi, G., Hangody, L., Modis, L., 2014. Mosaic arthroplasty of the medial femoral condyle in horses – An experimental study. Acta Veterinaria Hungarica 62, 155–168.
- Brittberg, M., 2008. Autologous chondrocyte implantation: Technique and long-term follow-up. Injury 39 (Suppl. 1), 40–49.
- Brittberg, M., Lindahl, A., Nilsson, A., Ohlsson, C., Isaksson, O., Peterson, L., 1994. Treatment of deep cartilage defects in the knee with autologous chondrocyte transplantation. The New England Journal of Medicine 331, 889–895.
- Bugbee, W.D., Pallante-Kichura, A.L., Görtz, S., Amiel, D., Sah, R., 2015. Osteochondral allograft transplantation in cartilage repair: Graft storage paradigm, translational models, and clinical applications. Journal of Orthopaedic Research 1–8. doi:10.1002/jor.22998.
- Changoor, A., Hurtig, M.B., Runciman, R.J., Quesnel, A.J., Dickey, J.P., Lowerison, M., 2006. Mapping of donor and recipient site properties for osteochondral graft reconstruction of subchondral cystic lesions in the equine stifle joint. Equine Veterinary Journal 38, 330–336.
- Chen, H., Sun, J., Hoemann, C.D., 2009. Drilling and microfracture lead to different bone structure and necrosis during bone-marrow stimulation for cartilage repair. Journal of Orthopaedic Research 27, 1432–1438.
- Chen, H., Chevrier, A., Hoemann, C.D., Sun, J., Ouyang, W., Buschmann, M.D., 2011. Characterization of subchondral bone repair for marrow-stimulated chondral defects and its relationship to articular cartilage resurfacing. The American Journal of Sports Medicine 39, 1731–1740.

- Cole, B., Fortier, L.A., Cook, J., Cross, J., Chapman, H., Roller, B., 2015. The use of micronized allograft articular cartilage (BioCartilage) and platelet-rich plasma to augment marrow stimulation in an equine model of articular cartilage defects. Arthroscopy The Journal of Arthroscopy and Related Surgery 31, E22. doi:10.1177/2325967115S00044.
- Convery, F.R., Akeson, W.H., Keown, G.H., 1972. The repair of large osteochondral defects. Clinical Orthopaedics and Related Research 82, 253–262.
- De Caro, F., Bisicchia, S., Amendola, A., Ding, L., 2015. Large fresh osteochondral allografts of the knee: A systematic clinical and basic science review of the literature. Arthroscopy: The Journal of Arthroscopic and Related Surgery 31, 757–765.
- De Schauwer, C., Meyer, E., Cornillie, P., De Vliegher, S., van de Walle, G.R., Hoogewijs, M., Declercq, H., Govaere, J., Demeyere, K., Cornelissen, M., et al., 2011. Optimization of the isolation, culture, and characterization of equine umbilical cord blood mesenchymal stromal cells. Tissue Engineering. Part C, Methods 17, 1061–1070.
- De Windt, T.S., Hendriks, J.A.A., Zhao, X., Vonk, L.A., Creemers, L.B., Dhert, W.J.A., Randolph, M.A., Saris, D.B.F., 2015. Concise review: Unraveling stem cell cocultures in regenerative medicine: Which cell interactions steer cartilage regeneration and how? Stem Cells Translational Medicine 3, 723–733.
- Desjardins, M.R., Hurtig, M.B., 1990. Cartilage healing: A review with emphasis on the equine model. Canadian Veterinary Journal 31, 565–572.
- Di Bella, C., Fosang, A., Donati, D.M., Wallace, G.G., Choong, P.F.M., 2015. 3D bioprinting of cartilage for orthopedic surgeons: Reading between the lines. Frontiers in Surgery 2, 1–7.
- Doral, M.N., Bilge, O., Batmaz, G., Donmez, G., Turhan, E., Demirel, M., Atay, O.A., Uzumcugil, A., Atesok, K., Kaya, D., 2012. Treatment of osteochondral lesions of the talus with microfracture technique and postoperative hyaluronan injection. Knee Surgery, Sports Traumatology, Arthroscopy 20, 1398–1403.Drobnic, M., Radosavljevic, D., Cör, A., Brittberg, M., Strazar, K., 2010. Debridement
- Drobnic, M., Radosavljevic, D., Cör, A., Brittberg, M., Strazar, K., 2010. Debridement of cartilage lesions before autologous chondrocyte implantation by open and arthroscopic techniques. The Journal of Bone and Joint Surgery 92B, 602– 608.
- Eldracher, M., Orth, P., Cucchiarini, M., Pape, D., Madry, H., 2014. Small subchondral drill holes improve marrow stimulation of articular cartilage defects. The American Journal of Sports Medicine 42, 2741–2750.
- Eltawil, N.M., Howie, S.E.M., Simpson, A.H.R.W., Amin, A.K., Hall, A.C., 2015. The use of hyperosmotic saline for chondroprotection: Implications for orthopaedic surgery and cartilage repair. Osteoarthritis and Cartilage 23, 469–477.
- Ferris, D.J., Frisbie, D.D., Kisiday, J.D., McIlwraith, C.W., Hague, B.A., Major, M.D., Schneider, R.K., Zubrod, C.J., Kawcak, C.E., Goodrich, L.R., 2014. Clinical follow-up of thirty-three horses treated for stifle injury with bone marrow-derived mesenchymal stem cells intra-articularly. Veterinary Surgery 43, 255–265.
- Ficat, R.P., Ficat, D., Gedeon, P., Toussaint, J.B., 1979. Spongialization: A new treatment for diseased patellae. Clinical Orthopaedics and Related Research 144, 74–83.
- Foland, J.W., McIlwraith, C.W., Trotter, G.W., 1992. Osteochondritis dissecans in the femoropatellar joint: Results of treatment with arthroscopic surgery. Equine Veterinary Journal 24, 419–423.
- Fortier, L.A., Nixon, A.J., Williams, J., Cable, C.S., 1998. Isolation and chondrocytic differentiation of equine bone marrow-derived mesenchymal stem cells. The American Journal of Veterinary Research 59, 1182–1187.
- Fortier, L.A., Mohammed, H.O., Lust, G., Nixon, A.J., 2002. Insulin-like growth factor-I enhances cell-based repair of articular cartilage. The Journal of Bone and Joint Surgery 84, 276–288.
- Fortier, L.A., Potter, H.G., Rickey, E.J., Schnabel, L.V., Foo, L.F., Chong, L.R., Stokol, T., Cheetham, J., Nixon, A.J., 2010. Concentrated bone marrow aspirate improves full-thickness cartilage repair compared with microfracture in the equine model. The Journal of Bone and Joint Surgery 92, 1927–1937.
- Frisbie, D.D., Stewart, M.C., 2011. Cell-based therapies for equine joint disease. The Veterinary Clinics of North America. Equine Practice 27, 335–349.
- Frisbie, D.D., Trotter, G.W., Powers, B.E., Rodkey, W.G., Steadman, J.R., Howard, R.D., Park, R.D., Mcllwraith, C.W., 1999. Arthroscopic subchondral plate microfracture technique augments healing of large chondral defects in the radial carpal bone and medial femoral condyle of horses. Veterinary Surgery 28, 242–255.
- Frisbie, D.D., Oxford, J.T., Southwood, L., Trotter, G.W., Rodkey, W.G., Steadman, J.R., Goodnight, J.L., McIlwraith, C.W., 2003. Early events in cartilage repair after subchondral bone microfracture. Clinical Orthopaedics and Related Research 407, 215–227.
- Frisbie, D.D., Morisset, S., Ho, C.P., Rodkey, W.G., Steadman, J.R., McIlwraith, C.W., 2006. Effects of calcified cartilage on healing of chondral defects treated with microfracture in horses. The American Journal of Sports Medicine 11, 1824–1831.
- Frisbie, D.D., Bowman, S.M., Colhoun, H.A., DiCarlo, E.F., Kawcak, C.E., McIlwraith, C.W., 2008. Evaluation of autologous chondrocyte transplantation via a collagen membrane in equine articular cartilage defects – Results at 12 and 18 months. Osteoarthritis and Cartilage 16, 667–679.
- Frisbie, D.D., Kisiday, J.D., Kawcak, C.E., Werpy, N.M., McIlwraith, C.W., 2009. Evaluation of adipose-derived stromal vascular fraction or bone marrow-derived mesenchymal stem cells for treatment of osteoarthritis. Journal of Orthopaedic Research 27, 1675–1680.
- Frisbie, D.D., McCarthy, H.E., Archer, C.W., Barrett, M.F., McIlwraith, C.W., 2015. Evaluation of articular cartilage progenitor cells for the repair of articular defects in an equine model. The Journal of Bone and Joint Surgery 97, 484–493.
- Gobbi, A., Chaurasia, S., Karnatzikos, G., Nakamura, N., 2015. Matrix-induced autologous chondrocyte implantation versus multipotent stem cells for the treatment of large patellofemoral chondral lesions: A nonrandomized prospective trial. Cartilage 6, 82–97.

Gomoll, A.H., Minas, T., 2014. The quality of healing: Articular cartilage. Wound Repair and Regeneration 22, 30–38.

- Goodrich, L.R., Chen, A., Werpy, N.M., Kisiday, J.D., Morley, P., McIlwraith, C.W., Sah, R.L., Chu, C., 2013. Autologous platelet enhanced fibrin (APEF) scaffold supports in situ repair in an equine model. In: Proceedings of the 11th Annual Meeting of the International Cartilage Research Society, Izmir, Turkey, 11.4.8.
- Goyal, D., Keyhani, S., Lee, E.H., Hui, J.H., 2013. Evidence-based status of microfracture technique: A systematic review of level I and II studies. Arthroscopy: The Journal of Arthroscopic and Related Surgery 29, 1579–1588.
- Griffin, D.J., Bonnevie, E.D., Lachowsky, D.J., Hart, J.C., Sparks, H.D., Moran, N., Matthews, G., Nixon, A.J., Cohen, I., Bonassar, L.J., 2015. Mechanical characterization of matrix-induced autologous chondrocyte implantation (MACI®) grafts in an equine model at 53 weeks. Journal of Biomechanics 48, 1944–1949. doi:10.1016/j.jbiomech.2015.04.010.
- Groll, J., Boland, T., Burdick, J., Blunk, T., Cho, D.-W., Dalton, P.D., Derby, B., Forgacs, G., Li, Q., Mironov, V.A., et al., 2016. Biofabrication: An updated definition for a field that is outgrowing its infancy. Biofabrication 8, 013001.
- Gutierrez-Nibeyro, S.D., 2011. Commercial cell-based therapies for musculoskeletal injuries in horses. Veterinary Clinics of North America, Equine Practice 27, 363–371.
- Hangody, L., Fules, P., 2003. Autologous osteochondral mosaicplasty for the treatment of full-thickness defects of the weight-bearing joints: Ten years of experimental and clinical experience. The Journal of Bone and Joint Surgery 85A, 25–32.
- Hangody, L., Kish, G., Karpati, Z., Szerb, I., Udvarhelyi, I., 1997. Arthroscopic autogenous osteochondral mosaicplasty for the treatment of femoral condylar articular defects. A preliminary report. Knee Surgery, Sports Traumatology, Arthroscopy 5, 262–267.
- Hendrickson, D.A., Nixon, A.J., Grande, D.A., Todhunter, R.J., Minor, R.M., Erb, H., Lust, G., 1994. Chondrocyte-fibrin matrix transplants for resurfacing extensive cartilage defects. Journal of Orthopaedic Research 12, 485–497.
- Hendrix, S.M., Baxter, G.M., McIlwraith, C.W., Hendrickson, D.A., Goodrich, L.R., Frisbie, D.D., Trotter, G.W., 2010. Concurrent or sequential development of medial meniscal and subchondral cystic lesions within the medial femorotibial joint in horses. Equine Veterinary Journal 42, 5–9.
- Hidaka, C., Goodrich, L.R., Chen, C.-T., Warren, R.F., Crystal, R.G., Nixon, A.J., 2003. Acceleration of cartilage repair by genetically modified chondrocytes overexpressing bone morphogenetic protein-7. Journal of Orthopaedic Research 21, 573–583.
- Hopper, N., Henson, F., Brooks, R., Ali, E., Rushton, N., Wardale, J., 2015a. Peripheral blood derived mononuclear cells enhance osteoarthritic human chondrocyte migration. Arthritis Research and Therapy 17, 1–10.
- Hopper, N., Wardale, J., Brooks, R., Power, J., Rushton, N., Henson, F., 2015b. Peripheral blood mononuclear cells enhance cartilage repair in in vivo osteochondral defect model. PLoS ONE 10, e0133937.
- Hopper, N., Wardale, J., Howard, D., Brooks, R., Rushton, N., Henson, F., 2015c. Peripheral blood derived mononuclear cells enhance migration and chondrogenic differentiation of multipotent mesenchymal stromal cells. Stem Cells International 2015, 323454.
- Howard, R.D., McIlwraith, C.W., Trotter, G.W., 1995. Arthroscopic surgery of subchondral cystic lesions of the medial femoral condyle in horses: 41 cases (1988–1991). The Journal of American Veterinary Medical Association 206, 842–850.
- Hunter, W., 1743. On the structure and diseases of articular cartilages. Philosophical Transactions of the Royal Society of London 42B, 514–521.
- Hunziker, E.B., 1999. Articular cartilage repair: Are the intrinsic biological constraints undermining this process insuperable? Osteoarthritis and Cartilage 7, 15–28.
- Hunziker, E.B., 2002. Articular cartilage repair. Basic science and clinical progress. A review of the current status and prospect. Osteoarthritis and Cartilage 10, 432–463.
- Hunziker, E.B., Quin, T.M., 2003. Surgical removal of articular cartilage defects leads to loss of chondrocytes from cartilage bordering the wound edge. Journal of Bone and Joint Surgery 85 (Suppl. 2), 85–92.
 Hunziker, E.B., Rosenberg, L.C., 1996. Repair of partial-thickness defects in articular
- Hunziker, E.B., Rosenberg, L.C., 1996. Repair of partial-thickness defects in articular cartilage: Cell recruitment from the synovial membrane. The Journal of Bone and Joint Surgery 5, 721–733.
- Hunziker, E.B., Lippuner, K., Keel, M.J.B., Shintani, N., 2015. An educational review of cartilage repair: Precepts & practice – Myths & misconceptions – Progress & prospects. Osteoarthritis and Cartilage 23, 334–350.
- Hurtig, M.B., Fretz, P.B., Doige, C.E., Schnurr, D.L., 1988. Effects of lesion size and location of equine articular cartilage repair. Canadian Journal of Veterinary Research 52, 137–146.
- Hurtig, M.B., Pearce, S.G., Warren, S., 2001. Arthroscopic mosaic arthroplasty in the equine third carpal bone. Veterinary Surgery 30, 228–239.
- Janicek, J.C., Cook, J.L., Wilson, D.A., Ketzner, K.M., 2010. Multiple osteochondral autografts for treatment of a medial trochlear ridge subchondral cystic lesion in the equine tarsus. Veterinary Surgery 39, 95–100.
- Jayasuriya, C.T., Chen, Q., 2015. Potential benefits and limitations of utilizing chondroprogenitors in cell-based cartilage therapy. Connective Tissue Research 56, 265–271.
- Karakaplan, M., Elmali, N., Mirel, E., Sahin, N., Ergen, E., 2015. Effect of microfracture and autologous conditioned plasma application in the focal full-thickness chondral defect of the knee: An experimental study on rabbits. The Journal of Orthopaedic Surgery and Research 10, 1–7.Kisiday, J.D., Goodrich, L.R., McIlwraith, C.W., Frisbie, D.D., 2013. Effects of equine
- Kisiday, J.D., Goodrich, L.R., McIlwraith, C.W., Frisbie, D.D., 2013. Effects of equine bone marrow aspirate volume on isolation, proliferation, and differentiation potential of mesenchymal stem cells. The American Journal of Veterinary Research 74, 801–807.

- Klein, T.J., Malda, J., Hutmacher, D.W., 2009. Tissue engineering of articular cartilage with biomimetic zones. Tissue Engineering. Part B, Reviews 15, 143.
- Klontzas, M.E., Kenanidis, E.I., Heliotis, M., Tsiridis, E., Mantalaris, A., 2015. Bone and cartilage regeneration with the use of umbilical cord mesenchymal stem cells. Expert Opinion on Biological Therapy 15, 1541–1552.
- Koch, T.G., Heerkens, T., Thomsen, P.D., Betts, D.H., 2007. Isolation of mesenchymal stem cells from equine umbilical cord blood. Biomed Central Biotechnology 7, 1–9.
- Kon, E., Filardo, G., Di Matteo, B., Perdisa, F., Marcacci, M., 2013. Matrix assisted autologous chondrocyte transplantation for cartilage treatment: A systematic review. Bone and Joint Research 2, 18–25.
- Kon, E., Roffi, A., Filardo, G., Tesei, G., Marcacci, M., 2015. Scaffold-based cartilage treatments: With or without cells? A systematic review of preclinical and clinical evidence. Arthroscopy: The Journal of Arthroscopic and Related Surgery 31, 767–775.
- Kreuz, P.C., Steinwachs, M.R., Erggelet, C., Krause, S.J., Konrad, G., Uhl, M., Südkamp, N., 2006. Results after microfracture of full-thickness chondral defects in different compartments in the knee. Osteoarthritis and Cartilage 14, 1119–1125.
- Lee, H., Kirkland, W.G., Whitmore, R.N., Theis, K.M., Young, H.E., Richardson, A.J., Jackson, R.L., Hanson, R.R., 2014. Comparison of equine articular cartilage thickness in various joints. Connective Tissue Research 55, 339–347.
- Levato, R., Visser, J., Planell, J.A., Engel, E., Malda, J., Mateos-Timoneda, M.A., 2014. Biofabrication of tissue constructs by 3D bioprinting of cell-laden microcarriers. Biofabrication 6, 035020.
- Levett, P.A., Melchels, F.P.W., Schrobback, K., Hutmacher, D.W., Malda, J., Klein, T.J., 2014. Development of an artificial extracellular matrix of cartilage tissue engineering based on photo-curable gelatin, hyaluronic acid and chondroitin sulfate. Acta Biomaterialia 10, 214–223.
- Litzke, L.F., Wagner, E., Baumgartner, W., Hetzel, U., Josimovic-Alasevic, O., Libera, J., 2004. Repair of extensive articular cartilage defects in horses by autologous chondrocyte transplantation. Annals of Biomedical Engineering 32, 57–69.
- Lohan, P., Treacy, O., Lynch, K., Barry, F., Murphy, M., Griffin, M.D., Ritter, T., Ryan, A.E., 2015. Culture expanded primary chondrocytes have potent immunomodulatory properties and do not induce an allogeneic immune response. Osteoarthritis and Cartilage 1–13. doi:10.1016/j.joca.2015.10.005.
- Lubowitz, M.D., 2006. Partial-thickness articular cartilage defects: Evaluation and treatment. Operative Techniques in Orthopaedics 16, 227–231.
- Lynch, T.S., Patel, R.M., Benedick, A., Amin, N.H., Jones, M.H., Miniaci, A., 2015. Systematic review of autogenous osteochondral transplant outcomes. Arthroscopy: The Journal of Arthroscopic and Related Surgery 31, 746–754.
- Madeira, C., Santhagunam, A., Salgueiro, J.B., Cabral, J.M.S., 2015. Advanced cell therapies for articular cartilage regeneration. Trends in Biotechnology 33, 35–42.
- Makris, E.A., Gomoll, A.H., Konstantinos, N.M., Hu, J.C., Athanasiou, K.A., 2015. Repair and tissue engineering techniques for articular cartilage. Nature Reviews Rheumatology 11, 21–34.
- Malda, J., Benders, K.E.M., Klein, T.J., de Grauw, J.C., Kik, M.J.L., Hutmacher, D.W., Saris, D.B.F., van Weeren, P.R., Dhert, W.J.A., 2012. Comparative study of depthdependent characteristics of equine and human osteochondral tissue from the medial and lateral femoral condyles. Osteoarthritis and Cartilage 20, 1147–1151.
- Malda, J., Boere, J., van de Lest, C.H.A., van Weeren, P.R., Wauben, M.H.M., 2016. Extracellular vesicles for joint repair and regeneration. Nature Reviews Rheumatology 1–7. doi: 10.1038/nrrheum.2015.170.
- Mankin, H.J., 1974. The reactions of articular cartilage to injury and osteoarthritis. The New England Journal of Medicine 291, 1285–1292.
- Marmotti, A., Rossi, R., Castoldi, F., Roveda, E., Michielon, G., Peretti, G.M., 2015. PRP and articular cartilage: A clinical update. BioMed Research International 2015, 542502.
- Marquez-Curtis, L.A., Janowska-Wieczorek, A., McGann, L.E., Elliott, J.A.W., 2015. Mesenchymal stromal cells derived from various tissues: Biological, clinical and cryopreservation aspects. Cryobiology 71, 181–197.
- Martins, E.A.N., Michelacci, Y.M., Baccarin, R.Y.A., Cogliati, B., Silva, L.C.L.C., 2014. Evaluation of chitosan-GP hydrogel biocompatibility in osteochondral defects: An experimental approach. Biomed Central Veterinary Research 10, 1–9.
- McCarthy, H.E., Bara, J.J., Brakspear, K., Singhrao, S.K., Archer, C.W., 2012. The comparison of equine articular cartilage progenitor cells and bone marrowderived stromal cells as potential cell sources for cartilage repair in the horse. The Veterinary Journal 192, 345–351.
- McIlwraith, C.W., 2013. Surgical versus conservative management of osteochondrosis. The Veterinary Journal 197, 19–28.
- McIlwraith, C.W., Yovich, J.V., Martin, G.S., 1987. Arthroscopic surgery for the treatment of osteochondral chip fractures in the equine carpus. Journal of American Veterinary Medical Association 191, 531–540.
- McIlwraith, C.W., Foerner, J.J., Davis, D.M., 1991. Osteochondritis dissecans of the tarsocrural joint: Results of treatment with arthroscopic surgery. Equine Veterinary Journal 23, 155–162.
- McIlwraith, C.W., Nixon, A.J., Wright, I.M., Boening, K.J., 2005. Arthroscopic methods for cartilage repair. In: Diagnostic and Surgical Arthroscopy in the Horse, vol. 17, Third Ed. Mosby Elsevier, pp. 455–742.
- McIlwraith, C.W., Fortier, L.A., Frisbie, D.D., Nixon, A.J., 2011a. Equine models of articular cartilage repair. Cartilage 2, 317–326.
- McIlwraith, C.W., Frisbie, D.D., Rodkey, W.G., Kisiday, J.D., Werpy, N.M., Kawcak, C.E., Steadman, J.R., 2011b. Evaluation of intra-articular mesenchymal stem cells to augment healing of microfractured chondral defects. Arthroscopy: The Journal of Arthroscopic and Related Surgery 27, 1552–1561.

McIlwraith, C.W., Nixon, A.J., Wright, I.M., 2014. Arthroscopic surgery of the femoropatellar joint. In: Diagnostic and Surgical Arthroscopy in the Horse, Fourth Ed. Elsevier, Philadelphia, PA, pp. 186–209.

Mirza, M.Z., Swenson, R.D., Lynch, S.A., 2015. Knee cartilage defect: marrow stimulating techniques. Current Reviews in Musculoskeletal Medicine 8, 451–456.

Mithoefer, K., McAdams, T., Williams, R.J., 3rd, Kreuz, P.C., Mandelbaum, B.R., 2009. Clinical efficacy of microfracture technique for articular cartilage repair in the knee: An evidence-based systematic analysis. The American Journal of Sports Medicine 37, 2053–2063.

Moreira Teixeira, L.S., Patterson, J., Luyten, F.P., 2014. Skeletal tissue regeneration: Where can hydrogels play a role? International Orthopaedics (SICOT) 38, 1861–1876.

Nixon, A.J., Lust, G., Vernier-Singer, M., 1992. Isolation, propagation, and cryopreservation of equine articular chondrocytes. The American Journal of Veterinary Research 53, 2364–2370.

Nixon, A.J., Fortier, L.A., Goodrich, L.R., Ducharme, N.G., 2004. Arthroscopic reattachment of osteochondritis dissecans lesions using resorbable polydioxanone pins. Equine Veterinary Journal 36, 376–383.

Nixon, A.J., Begum, L., Mohammed, H.O., Huibregtse, B., O'Callaghan, M.M., Matthews, G.L., 2011. Autologous chondrocyte implantation drives early chondrogenesis and organized repair in extensive full- and partial-thickness cartilage defects in an equine model. Journal of Orthopaedic Research 29, 1121–1130.

Nixon, A.J., Rickey, E., Butler, T.J., Scimeca, M.S., Moran, N., Matthews, G.L., 2015. A chondrocyte infiltrated collagen type I/III membrane (MACI® implant) improves cartilage healing in the equine patellofemoral joint model. Osteoarthritis and Cartilage 23, 648–660.

Orth, P., Goebel, L., Wolfram, U., Ong, M.F., Gräber, S., Kohn, D., Cucchiarini, M., Ignatius, A., Pape, D., Madry, H., 2012. Effect of subchondral drilling on the microarchitecture of subchondral bone: Analysis in a large animal model at 6 months. The American Journal of Sports Medicine 40, 828–836.

Orth, P., Duffner, J., Zurakowski, D., Cucchiarini, M., Madry, H., 2015. Small-diameter awls improve articular cartilage repair after microfracture treatment in a translational animal model. The American Journal of Sports Medicine 1–11. doi:10.1177/0363546515610507.

Ortved, K.F., Nixon, A.J., Mohammed, H.O., Fortier, L.A., 2012. Treatment of subchondral cystic lesions of the medial femoral condyle of mature horses with growth factor enhanced chondrocyte grafts: A retrospective study of 49 cases. Equine Veterinary Journal 44, 606–613.

Ortved, K.F., Begum, L., Mohammed, H.O., Nixon, A.J., 2014. Implantation of rAAV5-IGF-I transduced autologous chondrocytes improves cartilage repair in fullthickness defects in the equine model. Molecular Therapy 1–11.

Pallante, A.L., Chen, A.C., Ball, S.T., Amiel, D., Masuda, K., Sah, R.L., Bugbee, W.D., 2012a. The in vivo performance of osteochondral allografts in the goat is diminished with extended storage and decreased cartilage cellularity. The American Journal of Sports Medicine 40, 1814–1823.

Pallante, A.L., Görtz, S., Chen, A.C., Healey, R.M., Chase, D.C., Ball, S.T., Amiel, D., Sah, R.L., Bugbee, W.D., 2012b. Treatment of articular cartilage defects in the goat with frozen versus fresh osteochondral allografts: Effects on cartilage stiffness, zonal composition, and structure at six months. The Journal of Bone and Joint Surgery 94, 1984–1995.

Peterson, L., Vasiliadis, H.S., Brittberg, M., Lindahl, A., 2010. Autologous chondrocyte transplantation. A long-term follow up. The American Journal of Sports Medicine 38, 1117–1124.

Pezzanite, L.M., Fortier, L.A., Antczak, D.F., Cassano, J.M., Brosnahan, M.M., Miller, D., Schnabel, L.V., 2015. Equine allogeneic bone marrow-derived mesenchymal stromal cells elicit antibody responses in vivo. Stem Cell Research and Therapy 6, 1–11.

Pigott, J.H., Ishihara, A., Wellman, M.L., Russell, D.S., Bertone, A.L., 2013. Investigation of the immune response to autologous, allogeneic, and xenogeneic mesenchymal stem cells after intra-articular injection in horses. Veterinary Immunology and Immunopathology 156, 99–106.

Prado, A.A.F., Favaron, P.O., Lopes Correia da Silva, L.C., Baccarin, R.Y.A., Miglino, M.A., Maria, D.A., 2015. Characterization of mesenchymal stem cells derived from the equine synovial fluid and membrane. Biomed Central Veterinary Research 11, 1–13.

Pridie, K.H., 1959. A method of resurfacing ostheoarthritic knee joints: Proceedings of the British Orthopedic Association. The Journal of Bone and Joint Surgery 41, 618–619.

Riboh, J.C., Cole, B.J., Farr, J., 2015. Particulated articular cartilage for symptomatic chondral defects of the knee. Current Reviews in Musculoskeletal Medicine 8, 429–435.

Rudd, R.G., Visco, D.M., Kincaid, S.A., Cantwell, H.D., 1987. The effects of beveling the margins of articular cartilage defects in immature dogs. Veterinary Surgery 16, 378–383.

Ryan, A., Bertone, A.L., Kaeding, C.C., Backstrom, K.C., Weisbrode, S.E., 2003. The effects of radiofrequency energy treatment on chondrocytes and matrix of fibrillated articular cartilage. The American Journal of Sports Medicine 3, 386–391.

Sams, A.E., Nixon, A.J., 1995. Chondrocyte-laden collagen scaffolds for resurfacing extensive articular cartilage defects. Osteoarthritis and Cartilage 3, 47–59.

Sams, A.E., Minor, R.R., Wootton, J.A.M., Mohammed, H., Nixon, A.J., 1995. Local and remote matrix responses to chondrocyte-laden collagen scaffold implantation in extensive articular cartilage defects. Osteoarthritis and Cartilage 3, 61–70.

Samuels, J., Krasnokutsky, S., Abramson, S.B., 2008. Osteoarthritis. A tale of three tissues. Bulletin of the NYU Hospital for Joint Diseases 66, 244–250.

Sandler, E.A., Bramlage, L.R., Embertson, R.M., Ruggles, A.J., Frisbie, D.D., 2002. Correlation of lesion size with racing performance after arthroscopic surgical treatment of subchondral cystic lesions of the medial femoral condyle: 150 cases (1989–2000). In: Proceedings of the 48th AAEP, Orlando, USA, pp. 255–256.

- Schnabel, L.V., Fortier, L.A., McIlwraith, C.W., Nobert, K.M., 2013. Review. Therapeutic use of stem cells in horses: Which type, how, and when? The Veterinary Journal 197, 570–577.
- Schuurman, W., Khristov, V., Pot, M.W., van Weeren, P.R., Dhert, W.J.A., Malda, J., 2011. Bioprinting of hybrid tissue constructs with tailorable mechanical properties. Biofabrication 3, 1–7.
- Schuurman, W., Levett, P.A., Pot, M.W., van Weeren, P.R., Dhert, W.J., Hutmacher, D.W., Mechels, F.P., Klein, T.J., Malda, J., 2013. Gelatin-methacrylamide hydrogels as potential biomaterials for fabrication of tissue-engineered cartilage constructs. Macromolecular Bioscience 13, 551–561.

Scott, G.S.P., Crawford, W.H., Colahan, P.T., 2004. Arthroscopic findings in horses with subtle radiographic evidence of osteochondral lesions of the medial femoral condyle: 15 cases (1995–2002). The Journal of American Veterinary Medical Association 224, 1821–1826.

Seo, J., Kambayashi, Y., Megumo, I., Haneda, S., Yamada, K., Furuoka, H., Tabata, Y., Sasaki, N., 2015a. Effects of a synovial flap and gelatin/β-tricalcium phosphate sponges loaded with mesenchymal stem cells, bone morphogenetic protein-2, and platelet rich plasma on equine osteochondral defects. Research in Veterinary Science 101, 140–143.

Seo, S.J., Mahapatra, C., Singh, R.K., Knowles, J.C., Kim, H.W., 2015b. Strategies for osteochondral repair: Focus on scaffolds. Journal of Tissue Engineering 5, 1–14.

Shamis, L.D., Bramlage, L.R., Gabel, A.A., Weisbrode, S., 1989. Effect of subchondral drilling on repair of partial-thickness cartilage defects of third carpal bones in horses. The American Journal of Veterinary Research 50, 290–295.

Sherman, S.L., Garrity, J., Bauer, K., Cook, J., Stannard, J., Bugbee, W., 2014. Fresh osteochondral allograft transplantation for the knee: Current concepts. The Journal of American Academy of Orthopaedic Surgeons 22, 121–133.

Smillie, I.S., 1957. Treatment of osteochondritis dissecans. The Journal of Bone and Joint Surgery 39, 248–260.

Smith, B., Sigal, I.R., Grande, D.A., 2015. Immunology and cartilage regeneration. Immunology Research 63, 181–186.

Smith, B.D., Grande, D.A., 2015. The current state of scaffolds for musculoskeletal regenerative applications. Nature Reviews Rheumatology 11, 213–222.

- Smith, M.A., Walmsley, J.P., Philips, T.J., Pinchbeck, G.L., Booth, T.M., Greet, T.R.C., Richardson, D.W., Ross, M.W., Schramme, M.C., Singer, E.R., et al., 2005. Effect of age at presentation on outcome following arthroscopic debridement of subchondral cystic lesions of the medial femoral condyle: 85 horses (1993–2003). Equine Veterinary Journal 37, 175–180.
- Spaas, J.H., Broeckx, S.Y., Chiers, K., Ferguson, S.J., Casarosa, M., Van Bruaene, N., Forsyth, R., Duchateau, L., Franco-Obregòn, A., Wuertz-Kozak, K., 2015. Chondrogenic priming at reduced cell density enhances cartilage adhesion of equine allogeneic MSCs – a loading sensitive phenomenon in an organ culture study with 180 explants. Cellular Physiology and Biochemistry 37, 651– 665.
- Sparks, H.D., Nixon, A.J., Bogenrief, D.S., 2011a. Reattachment of the articular cartilage component of type 1 subchondral cystic lesions of the medial femoral condyle with polydioxanone pins in 3 horses. Journal of American Veterinary Medical Association 238, 636–640.
- Sparks, H.D., Nixon, A.J., Fortier, L.A., Mohammed, H.O., 2011b. Arthroscopic reattachment of osteochondritis dissecans cartilage flaps of the femoropatellar joint: Long-term results. Equine Veterinary Journal 43, 650–659.

Spiller, K.L., Maher, S.A., Lowman, A.M., 2011. Hydrogels for the repair of articular cartilage defects. Tissue Engineering. Part B, Reviews 17, 281–299.

Stanish, W.D., McCormack, R., Forriol, F., Mohtadi, N., Pelet, S., Desnoyers, J., Restrepo, A., Shive, M.S., 2013. Novel Scaffold-Based BST-Cargel treatment results in superior cartilage repair compared to microfracture in a randomized controlled trial. The Journal of Bone and Joint Surgery 95, 1640–1650.

Steadman, J.R., Rodkey, W.G., Singleton, S.B., Briggs, K.K., 1997. Microfracture technique for full-thickness chondral defects: Technique and clinical results. Operative Techniques in Orthopaedics 7, 300–304.

Steadman, J.R., Rodkey, W.G., Briggs, K.K., 2010. Microfracture: Its history and experience of the developing surgeon. Cartilage 1, 78–86.

Taylor, S.E., Clegg, P.D., 2011. Collection and propagation methods for mesenchymal stromal cells. The Veterinary Clinics of North America. Equine Practice 27, 263–274.

UpRichard, K., Elce, Y.A., Piat, P., Beauchamp, G., Laverty, S., 2013. Outcome after arthroscopic treatment of lateral femoral trochlear ridge osteochondrosis in sport horses. Veterinary and Comparative Orthopaedics and Traumatology 2, 105– 109.

Vachon, A., Bramlage, L.R., Gabel, A.A., Weisbrode, S., 1986. Evaluation of the repair process of cartilage defects of the equine third carpal bone with and without subchondral bone perforation. The American Journal of Veterinary Research 47, 2637–2645.

Vilela, C.A., Correia, C., Oliveira, J.M., Sousa, R.A., Espregueria-Mendes, J., Reis, R.L., 2015. Cartilage repair using hydrogels: A critical review of in vivo experimental designs. ACS Biomaterials Science & Engineering 1, 726–739.

Visser, J., Peters, B., Burger, T.J., Boomstra, J., Dhert, W.J.A., Melchels, F.P.W., Malda, J., 2013. Biofabrication of multi-material anatomically shaped tissue constructs. Biofabrication 5, 035007.

Visser, J., Melchels, F.P., Jeon, J.E., van Bussel, E.M., Kimpton, L.S., Byrne, H.M., Dhert, W.J., Dalton, P.D., Hutmacher, D.W., Malda, J., 2015. Reinforcement of hydrogels using three-dimensionally printed microfibres. Nature Communications 6, 6933.

Wang, H., Zhou, L., Liao, J., Tan, Y., Ouyang, K., Ning, C., Ni, G., Tan, G., 2014. Cell-laden photocrosslinked GelMA – DexMA copolymer hydrogels with tunable mechanical properties for tissue engineering. Journal of Material Science: Materials in Medicine 25, 2173–2183.

- Wilke, M.M., Nydam, D.V., Nixon, A.J., 2007. Enhanced early chondrogenesis in articular cartilage defects following arthroscopic mesenchymal stem cell implantation in an equine model. Journal of Orthopaedic Research 25, 913–925.
- Williams, L.B., Koenig, J.B., Black, B., Gibson, T.W.G., Sharif, S., Koch, T.G., 2015. Equine allogeneic umbilical cord blood derived mesenchymal stromal cells reduce synovial fluid nucleated cell count and induce mild self-limiting inflammation when evaluated in an LPS induced synovitis model. Equine Veterinary Journal 1–7. doi:10.1111/evj.12477.
- Wu, L., Leijten, J.C.H., Georgi, N., Post, J.N., van Blitterswijk, C.A., Karperien, M., 2011. Trophic effects of mesenchymal stem cells increase chondrocyte proliferation and matrix formation. Tissue Engineering. Part A 17, 1425–1436.
- Yanke, A.B., Tilton, A.K., Wetters, N.G., Merkow, D.B., Cole, B.J., 2015. DeNovo NT particulated juvenile cartilage implant. Sports Medicine and Arthroscopy Review 23, 125–129.
- Zedde, P., Giachetti, G., Del Prete, F., Brunetti, A., Manunta, A.F., 2015. Remodeling of subchondral bone after nanofracture: an *in-vivo* ovine study. In: Proceedings of the 12th Annual Meeting of the International Cartilage Research Society, Chicago, USA, 8.4.5.